Author’s response to reviews

Title: Neuromyotonia with polyneuropathy, with prominent psychoorganic syndrome, insomnia, suicidal behavior, and without antibodies: a case report

Authors:

Edvard Ehler (eda.ehler@tiscali.cz)
Alena Meleková (alena.melekova@nemocnice-pardubice.cz)

Version: 6 Date: 17 March 2015

Author’s response to reviews: see over
Neuromyotonia with polyneuropathy, with prominent psychoorganic syndrome, insomnia, and suicidal behaviour, without antibodies. A case report.

Edvard Ehler, M.D., Ph.D., Alena Meleková, M.D.,

Thank You for important comments. We have changed our case report and in the abstract, beginning of discussion and in the conclusion we implemented the formulation – We describe here a case of xx, what is unique because of… And we tried to name some most important facts.
Changes are in red.

Sincerely,

Edvard Ehler, M.D., Ph.D.

17.3.2015